Author's response to reviews

Title: Telecare motivational interviewing for diabetes patient education and support: a randomised controlled trial based in primary care comparing nurse and peer supporter delivery

Authors:

Jeremy R Dale (Jeremy.Dale@warwick.ac.uk)
Isabela O Caramlau (Isabela.Caramlau@warwick.ac.uk)
Andrea Docherty (Andrea.Docherty@RedditchBromsgrove-pct.nhs.uk)
Jackie Sturt (Jackie.Sturt@warwick.ac.uk)
Hilary Hearnshaw (Hilary.Hearnshaw@warwick.ac.uk)

Version: 2 Date: 28 June 2007

Author's response to reviews: see over
Dear Dr. Altman,

RE: study protocol: Telecare motivational interviewing for diabetes patient education and support: a randomised controlled trial based in primary care comparing nurse and peer supporter delivery, MS: 1125308034133534

Thank you for confirming the acceptance of the above manuscript, pending formatting requirements. We have addressed them as follows:

1. Affiliations – they have been amended as required.
2. Superfluous text - we removed superfluous text from title page such as Title of article, Trials-.
3. Website addresses in text - we removed website addresses from the text on page 4. Also, we have shortened the website reference [29].
4. Ethics approval in Methods section – we have not made any changes, since it was already in the Methods section.
5. Authors’ contributions – we have added a sentence to confirm that all authors read and approved the final version of the manuscript – page 11

We have also made additional minor changes, after proof-reading the manuscript:

2. Reference [23] on page 9 was changed to [28].
3. Reference [28] on page 8 was changed to [25].
4. Page 9, last paragraph – the word ‘independent’ was deleted in the sentence ‘The data will be analysed independently by 2 independent researchers’.

If you have any further queries, please do not hesitate to contact me.

Kind regards,

Prof. Jeremy Dale
Health Sciences Research Institute
Warwick Medical School
University of Warwick
Coventry
CV4 7AL
Tel: 02476 522891
Fax: 02476 528375
Email: Jeremy.Dale@warwick.ac.uk